



Article Exploring the Quality of Dynamic Open Government Data Using Statistical and Machine Learning Methods

Areti Karamanou *[®], Petros Brimos[®], Evangelos Kalampokis[®] and Konstantinos Tarabanis[®]

Information Systems Lab, Department of Business Administration, University of Macedonia, 54636 Thessaloniki, Greece

* Correspondence: akarm@uom.edu.gr

Abstract: Dynamic data (including environmental, traffic, and sensor data) were recently recognized as an important part of Open Government Data (OGD). Although these data are of vital importance in the development of data intelligence applications, such as business applications that exploit traffic data to predict traffic demand, they are prone to data quality errors produced by, e.g., failures of sensors and network faults. This paper explores the quality of Dynamic Open Government Data. To that end, a single case is studied using traffic data from the official Greek OGD portal. The portal uses an Application Programming Interface (API), which is essential for effective dynamic data dissemination. Our research approach includes assessing data quality using statistical and machine learning methods to detect missing values and anomalies. Traffic flow-speed correlation analysis, seasonal-trend decomposition, and unsupervised isolation Forest (iForest) are used to detect anomalies. iForest anomalies are classified as sensor faults and unusual traffic conditions. The iForest algorithm is also trained on additional features, and the model is explained using explainable artificial intelligence. There are 20.16% missing traffic observations, and 50% of the sensors have 15.5% to 33.43% missing values. The average percent of anomalies per sensor is 71.1%, with only a few sensors having less than 10% anomalies. Seasonal-trend decomposition detected 12.6% anomalies in the data of these sensors, and iForest 11.6%, with very few overlaps. To the authors' knowledge, this is the first time a study has explored the quality of dynamic OGD.

Keywords: open government data; dynamic government data; high-valuable data; real-time data; traffic data; data quality; isolation forest; eXplainable artificial intelligence

1. Introduction

The open government data (OGD) movement, which urged governments and public organizations to open up their data for others to reuse for both private and commercial purposes, emerged at the beginning of the 21st century [1]. Innovative political initiatives such as the Public Sector Information (PSI) Directive in Europe in 2003 [2] and the U.S. President's Obama open data program in 2009 [3] supported this movement and made OGD a political priority. In the next years, numerous OGD portals at various administrative levels (e.g., local, regional, national, etc.) and in many countries across the globe were launched [1,4–6].

The expected benefits for citizens, businesses, and public administration and the potential impact on the society as a whole have been extensively described in the literature. OGD are expected to strengthen transparency [7] and improve decision-making processes [8], stimulate economic growth and innovation [9,10], and provide opportunities for the development of more effective public services [11,12], including Integrated Public Services (IPS) [13]. However, so far, the impact of OGD is still rather limited [14], with many studies exploring the reasons that hinder open government initiatives from reaching their full potential [6].



Citation: Karamanou, A.; Brimos, P.; Kalampokis, E.; Tarabanis, K. Exploring the Quality of Dynamic Open Government Data Using Statistical and Machine Learning Methods. *Sensors* **2022**, *22*, 9684. https://doi.org/10.3390/s22249684

Academic Editor: Amir H. Gandomi

Received: 2 November 2022 Accepted: 7 December 2022 Published: 10 December 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations.



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

OGD is a rapidly evolving phenomenon. The amount of OGD produced and disseminated increases exponentially, and new types of data are being generated. For example, dynamic data (including environmental, traffic, satellite, meteorological, and sensor-generated data) have been recently recognized by the European Commission as an important part of OGD, presenting huge potential economic value [15]. Due to the suitability of dynamic OGD for the creation of value-added services, applications and high-quality and decent jobs, they are recognized as high-value data (HVD), whose reuse has significant societal, environmental, and economic benefits [15]. The collection and dissemination of this type of data impose new requirements and challenges. For example, dynamic data are characterized by their high variability and rapid obsolescence, making their immediate availability and regular updates crucial for the creation of added-value services and applications. Moreover, sensors are prone to malfunctions caused by, e.g., bad weather and temperature conditions, resulting in anomalous observations [16]. As a result, anomaly detection and classification methods need to be applied to detect and correct corrupt or inaccurate records. Although the quality of OGD has been recognized as being of vital importance, related works (e.g., [6,17,18]) tend to focus more on the quality of the metadata and not on the data itself.

At the same time, innovative data analysis and exploitation methods for OGD emerge, such as artificial intelligence [19], including machine learning [20]. These technologies can provide opportunities for the creation of innovative and intelligent services and applications, which are built upon the use and combination of new types of OGD. These applications are considered "data intelligence applications". For example, data intelligence applications exploiting traffic data involve predicting various traffic states such as traffic flow, traffic speed, and traffic demand [21].

Hence, in order to be able to better understand and achieve the economic and social potential offered by the reuse of OGD, we need to keep pace with those rapid changes and investigate the exploitation of new types of OGD employing state-of-the-art technologies.

The objective of this paper is to explore the quality of Dynamic Open Government Data in order to facilitate the development of effective, efficient, and entrusted data intelligence applications. To this end, we focus on and study a single case, namely the traffic data of the Region of Attica that are provided through data.gov.gr (accessed on 24 October 2022), the official Greek OGD portal. This case was selected because it involves the use of an API for accessing HVD for traffic, which allows for immediate availability and regular updates of the data. Specifically, traffic data are updated hourly in the Greek OGD portal.

Our research approach involves the exploration and evaluation of the provided data regarding the existence of missing values and anomalies. Anomaly detection comprises both the identification of (i) anomalous flow-speed correlations and (ii) deviations from the normal traffic pattern (see Section 2.2 for details). These two present complementary views of the traffic data quality. Deviations from the normal traffic pattern are detected using a statistical method and an unsupervised machine learning method with one variable. Detected deviations are also classified in sensor faults and unusual traffic conditions. The quality of traffic data is further explored by implementing a second machine learning scenario that uses multiple variables.

This paper is organized as follows: Section 2 presents the background knowledge required to understand the content of this work. Section 3 describes in detail the steps followed in this research. Section 4 provides the details of the case vignettes of this work and explores traffic data. Then, Section 5 detects the missing values in the traffic data, while Section 6 detects anomalies in the data using three methods, namely flow-speed correlation analysis (Section 6.1), seasonal-trend decomposition using Loess (Section 6.2), and isolation forest (Section 6.3). It also classifies detected anomalies as sensor faults and unusual traffic conditions. Thereafter, Section 7 explores anomalies in the traffic data using multiple variables. Finally, Section 8 discusses the results of this work, and Section 9 concludes this study.

2. Background

This section presents the background knowledge required to understand the content of this paper. Specifically, it describes dynamic Open Government Data, including open traffic data, the anomaly detection methods used in this work, and anomaly explanation.

2.1. Dynamic Open Government Data

Open Government Data (OGD) are data published by the public sector, freely distributed to all citizens without further restrictions [22]. The term "open" refers to any data or information that can be equally and freely used, modified or shared by all citizens and businesses [11,23]. Today, a large number of public authorities and National Statistical Institutes internationally have already demonstrated their commitment to opening their data by launching OGD portals (e.g., the European Data Portal (https://data.europa.eu/ accessed on 25 October 2022)). OGD portals provide a plethora of datasets. For instance, the European Data Portal provides 1,538,031 datasets describing data that can be classified into 13 themes, including transport, economy and finance, environment, and population and society. In total, 0.75% of the datasets (10,667) come from Greece.

Dynamic data (including environmental, traffic, satellite, meteorological and sensorgenerated data) have been recently recognized as an important part of OGD [15]. The importance of dynamic data, also mentioned as real-time data, has already been stressed in the literature (e.g., [24]). Unlike static data, dynamic data are characterized by their high variability and rapid obsolescence, making their immediate availability and regular updates crucial for the creation of added-value services and applications. Hence, the transmission of dynamic data should be possible immediately after their collection via an Application Programming Interface (API).

Open Traffic Data

Dynamic (or real-time) data with traffic-related information (e.g., counted number of vehicles, average speed) generated from sensors are increasingly provided by Open Government Data (OGD) portals. This type of data facilitates the delivery of public services in the Smart City context, improving the quality of citizens' life and stimulating economic growth [25].

Open Traffic Data are usually aggregated by minute or hour and are available in various standard formats (e.g., using the JavaScript Object Notation—JSON, or eXtensible Markup Language, XML formats). However, that data are not always provided in real-time and cannot easily be retrieved to an external solution using an Application Programming Interface (API) [26]. Specifically, only a few of the OGD portals use an Application Programming Interface (API) to enable accessing and retrieving the data, hampering their use in data intelligence applications. In addition, only a few of the OGD portals provide streaming traffic data, i.e., updated traffic measurements every minute, while the rest of them update data every 5 min, 1 h or 1 day. Finally, some of them only provide access to historical data.

For instance, the Norwegian Public Roads Administrations' Traffic Data API (https: //www.vegvesen.no/trafikkdata/api accessed on 25 October 2022) provides hourly aggregated traffic data from the roads in Norway. The API can be accessed via a Graphical User Interface (GUI) or via programming languages to obtain data such as traffic volumes, spatial information, and number of lanes. The provided historical data trace back to 2019. The Swedish OGD portal (https://api.trafikinfo.trafikverket.se/ accessed on 25 October 2022) provides a GUI for querying streaming traffic data with the POST method for API. The data are returned in XML or JSON format and contain traffic flow data, arrivals and departures of ferries, road condition information, weather information etc. The streaming data are updated every minute. Historical data cannot be accessed through the API. In addition, the open API from Helsinki (https://helsinki-public.azurewebsites.net/ accessed on 25 October 2022) provides sensor traffic such as the volume of vehicles, their average speed and type. The open API is updated every 5 min and returns data from the past hour aggregated in 5 min, 1 h, or 1 day intervals in JSON format. Furthermore, a promising OGD initiative is provided by the government of Switzerland (https://opentransportdata.swiss/en/cookbook/rt-road-traffic-counters/ accessed on 25 October 2022), where various road authorities record traffic movements in a grouped, summarized dataset, which is freely accessible. The OGD traffic dataset contains traffic measures from the country's most important roads near significant urban regions and national highways. The dataset uses the DATEX II, a standard for the exchange of road traffic data based on a specific XML schema. The advantage of this portal is that it returns real-time streaming data every minute. New incoming data replace the last published data, while historical data are not available in this portal. To the authors' knowledge, the Swiss OGD portal is the only European open portal that provides streaming data using 1 min intervals. The data can be downloaded in JSON format with an http POST request using an API key (token) after a user's registration. Finally, this study examines the quality of sensor-generated data from the Greek OGD portal (https://www.data.gov.gr/datasets/road_traffic_attica/ accessed on 25 October 2022). Provided traffic data include hourly aggregated information for traffic flow and average speed can be downloaded in CSV or JSON format but can also be accessed via an API using authorization keys (tokens). Traffic data are updated hourly in the portal.

2.2. Anomaly Detection

Sensors are prone to malfunctions caused by, e.g., bad weather, temperature conditions, or damages, resulting in anomalous observations. There is a plethora of definitions in the data mining and statistics literature for anomalies. For instance, they have been defined as outliers, abnormalities, discordants, or deviants [27] and as observations that do not follow the well-defined normal behavior of data [28]. Under the assumption that the majority of observations in a dataset are normal, anomalies only represent a very small proportion of the dataset, and their distribution significantly deviates from the distribution of the rest of the data [29]. In this direction, anomaly detection refers to the research area and techniques that detect data points that deviate from the dataset by calculating the likelihood of this point being an anomaly, also called an anomaly score. Finally, a threshold value is defined so that points with an anomaly score greater than this threshold are considered anomalies. Therefore anomaly detection problems are defined as binary label classification tasks determining whether an observation is normal or an anomaly. Anomaly detection has a wide range of applications in several domains, such as intrusion detection for computerbased systems, fraud detection for banks and insurance companies, and medical anomaly detection for health monitoring systems [30]. In the Internet of Things (IoT) domain, sensor monitoring process demands high quality and trustworthy detection of corrupt sensor data, ensuring the quality of the transmitted data. In-situ sensors and wireless sensor networks (WSN) produce large sequences of observations. Therefore, anomaly detection for sensor-generated data is strongly linked with time-series analysis and forecasting.

There are two main categories for sensor anomaly detection: simple statistical or timeseries analysis techniques and machine learning techniques. The former refers to statistical approaches for time series anomaly detection such as regressive models ARIMA, moving average, HA [31] and time series decomposition techniques such as the seasonal-trend decomposition. The latter refers to machine learning methods that learn specific patterns from the time-series data and are further categorized into supervised and unsupervised approaches. Supervised anomaly detection methods are applied on labeled time series datasets, such that for each observation (timestamp), the label is known (anomaly or normal data point), and the dataset contains both normal points and outliers. Examples of supervised anomaly detection algorithms are neural networks LSTMs (Long Short Memory Networks) [32] and one-class SVM (Support Vector Machines) [33]. On the contrary, unsupervised anomaly detection methods assume that time-series data are unlabeled. Most of the unsupervised approaches use distance-based methods or auto-encoders (neural networks that reconstruct the input data) [34]. Finally, apart from statistical and machine learning methods, several domain-oriented thresholds are defined in order to determine whether an observation is considered anomalous. In our case, this threshold is defined as the flow-speed correlation value, the maximum number of vehicles that can pass a specific counting point (a sensor) in a particular time interval.

In our research approach, anomaly detection comprises both the identification of (a) anomalous flow-speed correlations and (b) deviations from the normal traffic pattern using a statistical time-series method and an unsupervised machine learning algorithm. These two present complementary views of the traffic data quality. To this end, we employ the following methods, respectively.

2.2.1. Anomalous Flow-Speed Correlation

In traffic data, the number of vehicles counted by a sensor and their average speed are strongly correlated. In particular, considering that each sensor measures data that pass from one or more lanes, the maximum number of vehicles that can pass in all lanes in one hour can be calculated as [35]:

$$number_of_vehicles = \frac{average_speed * 1000}{average_vehicle_length + \frac{average_speed}{3.6}} * number_of_lanes$$

where *average_speed* is the average speed provided by the sensors measured in km per hour and *average_vehicle_length* is the average length of the different types of vehicles, the fraction *average_speed*/3.6 represents the "safe driving distance" that should be kept between vehicles and is based on the vehicle speed, and *number_of_lanes* is the number of lanes in the road each sensor is positioned. The value of *average_vehicle_length* is set to four. When the number of vehicles measured by a sensor in an hour is higher than this value, then the measurement is considered an anomaly.

2.2.2. Seasonal-Trend Decomposition Using Loess for Anomaly Detection

The seasonal-trend decomposition of periodic time series using Loess (STL) is a fundamental method for time series analysis, with many applications in anomaly detection and forecasting [36]. The robust STL algorithm performs seasonal-trend decomposition using the statistical smoother "locally estimated scatter plot smoothing"-"Loess" (a generalization of the moving average technique) and locally-weighted regression functions to decompose the time series. Specifically, STL considers the original time series as a composition of three components (additive model):

$$y_t = T_t + S_t + R_t \tag{1}$$

where y_t is the observed data at time t, T_t denotes the trend in time series, S_t is the seasonal component of the original time series, and R_t denotes the remainder component. STL employs a statistical smoother called the loess (locally estimated scatter plot smoothing, a generalization of moving average technique), using locally-weighted regression functions to decompose the time series. The trend component shows a general pattern in time series on a long-term basis, the linear increasing (uptrend) or decreasing (downtrend). Furthermore, the seasonal component refers to the repeating patterns (periodic patterns) over time. Finally, the remaining variations in time series are the remainder component, also known as the noise. The remainder component is calculated by subtracting the trend and seasonal components from the original series. The remainder curve indicates the existence of noise present in the data. The decomposition procedure consists of the following steps, which are repeated iteratively until a final convergence [36]:

The first step is the noise removal from the time series using bilateral filtering, where neighbors with similar values are used to smooth the time series. The literature proposes a filter window of length 2H + 1 and filter weights computed by the Gaussian filter functions [37–39]. After denoising, the trend extraction is employed by modeling the trend difference using Least Absolute Deviations (LAD) and 11 regularizations. The trend is computed through the LAD loss function as an optimization problem by minimizing the

6 of 23

sum of absolute deviations (such as the least square technique). After the trend removal from the time series, the seasonal extraction is performed by a non-local seasonal filtering. The seasonality component is defined as a weighted linear combination of the 2H + 1 neighbors of y_t . The K-neighborhood technique for seasonality extraction indicates that data points with similar seasonality to y_t will be given larger weights based on their distance to y_t in the time dimension. As mentioned previously, these steps are performed iteratively by the STL algorithm until convergence. After the final iteration, the results of the STL decomposition algorithm are three time series and three independent graphs representing seasonal and residual trends.

STL decomposition is very useful for anomaly detection in time series by analyzing the residual curve of the STL output time series. For that reason, after the decomposition procedure, the remainder curve is divided into an area of normal data points and an area of outliers or anomalies. The limits of these areas in the remainder curve can be defined by various methods, such as the interquartile range (IQR) method or the empirical rule for normal distribution [40].

2.2.3. Unsupervised Anomaly Detection with Isolation Forest

The recent development of machine learning and Big data analytics has introduced a wide variety of methods for anomaly detection. Moreover, unsupervised learning methods seem to be significantly useful for anomaly detection problems due to their ability to learn patterns from unlabeled data. Since the majority of real-world datasets do not contain labeled anomalous data, unsupervised learning approaches are a suitable choice. Unsupervised learning models do not learn patterns on labeled data but perceive the data structure in order to classify it into a particular set of classes. In the anomaly detection context, they assume that only a very small proportion of the dataset is anomalous and data groups of similar instances are considered normal. Ref. [41] performs anomaly detection using the unsupervised approach for detecting anomalies on surveillance videos while [42] detects anomalies in hyperspectral images with an unsupervised Deep Belief Network. Furthermore, ref. [43] uses an unlabeled Satellite dataset to detect outliers, implementing a deep auto-encoder technique, and ref. [44] demonstrates a method based on unsupervised learning for detecting noise on acoustic sensors.

The isolation Forest (iForest) is an anomaly detection algorithm based on the hypothesis that outliers are always rare and a few data points among the whole dataset (far from the center of normal clusters) [45]. This ensemble method isolates the outlier data points from the rest of the data, by portioning the data set using isolation trees (binary trees). Each node of an isolation tree consists of a randomly selected attribute A and a split value S, such that S A, while S corresponds to the minimum or maximum value of the selected attribute. The general idea is that anomalous data points are very different and easier to divide than the normal data points, and they are also closer to the root nodes of the isolation trees [46,47]. After the creation of a completed forest by generating T random isolation trees, the path length h(x) to isolate a data point x is calculated using the average number of the path lengths to isolate x from each isolation tree. The anomaly score is defined as:

$$s(x) = 2^{-\frac{E[(h(x)]}{c(n)}},$$
 (2)

where c(n) is the average value of paths h(x) that is defined as:

$$c(n) = 2H(n-1) - \frac{2(n-1)}{n},$$
(3)

where *H* is the harmonic number, and it can be estimated by $\ln(i) + 0.5772156649$ (Euler's constant) [45]. If the anomaly score is closer to 1 the data point is considered an anomaly, *x* is an isolated point, and the path h(x) is relatively small. On the other hand, when h(x) is large, then *x* is not an isolated point and the anomaly score tends to 0; thus, the data point is considered nominal [45,48].

2.3. Anomaly Explanation

Machine learning models are complex algorithms, often referred as "black box", that in many cases do not provide enough explanations regarding their predictions so that human users can understand. During the last decade, there has been high concern among the research community about the lack of transparency, accountability and interpretability of complex machine learning predictive models that might have consequences in many domains [49]. Therefore, under the recent development of eXplainable Artificial Intelligence (XAI), a second post-hoc model is created that explicitly explains the first predictive "black box" model. XAI has been recently employed not only in Open Government Data (OGD) [20] but also in healthcare [50,51] and transport data [52]. This new field of XAI has applications in the anomaly detection research area named as eXplainable Anomaly Detection (XOD). Anomaly detection explanations refer to the definition of a post-hoc model that explains why the initial model categorizes the objects as normal or anomalies and determines the internal causes of their predictions. In particular, explanations in unsupervised anomaly detection are equivalent to a supervised classifier ignoring the training process, finding specific patterns and subspaces in the data, and performing statistical computations in order to determine the rank of data between inliers and outliers [53]. Post-hoc explanations are useful for highly unintelligible models such as neural networks and ensemble techniques (such as decision trees, random forests, and isolation forests).

The most commonly used methods developed to explain predictions (or anomalies) from supervised or unsupervised classifiers are LIME [54] and SHAP—SHapley Additive exPlanation [55]. LIME is an explainable model that is trained on a sample data, and the model's predictions use sparse linear models as explanations. LIME performs explanations by creating an interpretable model with local data around the neighborhood of an instance. After the model's predictions on each sample, these sample data are weighted to the adjacency to the given instance. For example, ref. [56] proposes an explainable anomaly detection framework using SHAP for predictive maintenance in a manufacturing system, and ref. [57] explains anomalies of a deep auto-encoder model with SHAP explanations. The SHAP framework is used by [58] to explain anomalies from maritime engines with the Isolation Forest algorithm. In this study, we implement the SHAP approach to explain the traffic anomalies from the OGD sensor dataset.

SHAP assigns an importance value, called the Shapley value, inspired by game theory, to explain a particular prediction in compliance with the following properties: (i) local accuracy—the explanation model matches the original model, (ii) missingness—missing features on the original model have no impact on the explanation and (iii) consistency—transformations of the model that increase or stabilize the contribution of a feature regardless of other inputs do not lead to the decrease in this input's attribution [55]. The prediction f(x) can be explained as:

$$f(x) = g(x') = \phi_0 + \phi_i x'_i$$
(4)

where g(x') is the explanation model, x is the input value, ϕ_0 is the baseline when all the input features are missing. According to [55], SHAP explains the output of an instance by calculating the Shapley value, i.e., the contribution of each feature to the output (prediction, anomaly):

$$\phi_i = \sum_{S \subseteq F/i} \frac{|S|!(|F| - |S| - 1)!}{|F|!} [f_{S \cup i}(x_{S \cup i}) - f_S(x_S)]$$
(5)

 ϕ_i denotes the importance of the ith variable, *F* is the set of all variables, *S* represents all possible subsets and $f_S(x_S)$, $f_{S\cup i}(x_{s\cup i})$ denote the output of the model before and after, including the *i*th variable to *S*, respectively.

3. Research Approach

The research approach of this work uses five steps, namely (1) data collection, (2) data exploration, (3) missing values analysis, (4) anomaly detection and classification, and (5) anomaly detection using multiple variables.

(1) Data collection. In this step, available traffic data from data.gov.gr (accessed on 25 October 2022) are collected using the data.gov.gr (accessed on 25 October 2022) API. These data have been produced by sensors that are positioned in the Attica Region in Greece. In addition, the position of the sensors is specified and mapped to latitude and longitude geographic coordinators in order to be able to present data in map visualizations.

(2) Data exploration. This step uses statistical analysis and visualizations to explore the collected traffic data for better understanding. Specifically, it explores the number of sensors that produced traffic data during each month and the InterQuartile Range (IQR) of the vehicles measured by each sensor. It also explores the correlation of the sensors. To this end, it employees the Pearson correlation coefficient for the number of vehicles counted each hour of the day by each sensor. The Pearson correlation is a commonly used measurement for searching correlations between time series, dividing the correlation among two variables for the product of the square of their variance. Sensors that are located nearby one another or three other sensors, i.e., their distance is lower than 2000 m, and that have a Pearson correlation coefficient greater than 0.8 are considered strongly correlated sensors. Weakly correlated sensors have a Pearson correlation coefficient greater than 0.7.

(3) Missing values analysis. Since the traffic data are dynamic data collected by sensors, there is a chance that some observations may be missing due to various reasons such as failures of sensors, network faults, and other issues. In this context, this step searches for observations that are missing from the traffic data based on two dimensions; (i) the time, and (ii) the sensors. The first case searches for the missing observations per day. To this end, we calculate for each day the number of observations that should be available for all sensors, and then we subtract this number from the number of available observations. This analysis was used to select the "best" time window for the traffic data, i.e., the time window with the least missing observations. For the second case, the total number of missing values per sensor is calculated. Additional statistical analyses are employed in order to explore the distribution of the sensors' missing observations.

(4) Anomaly detection and classification. This step identifies traffic data anomalies, i.e., observations that do not follow the well-defined normal behavior. It also classifies detected anomalies into (i) anomalies from unusual traffic conditions (e.g., traffic accidents, weather conditions), and (ii) anomalies from sensor faults. In this work, anomaly detection comprises both the identification of (a) anomalous flow-speed correlations using flow-speed correlation analysis, and (b) deviations from the normal traffic pattern using two methods, namely, Seasonal-Trend decomposition using Loess (STL), and unsupervised learning isolation Forest (iForest), which are described in Section 2. All methods are applied to traffic data regarding the time period described by the "best" time window found in the previous step. The flow-speed correlation analysis is used to calculate the number of anomalies per sensor. Descriptive statistics are then used to describe and summarize calculated anomalies. The flow-speed correlation analysis helps to identify the "best" sensors, i.e., sensors with the lowest number of anomalies. In order to further explore the quality of data, we use the STL and iForest methods to detect anomalies in data related to the "best" sensors. For these two methods, missing values in data are imputed using the linear interpolation method. Specifically, we apply the STL decomposition in the traffic data and examine the residual curve to detect anomalies. We then specify the upper and lower fence for outlier detection by applying the IQR method in the residual curve. We experiment with multiple values in order to find the best scalar that the IQR will be multiplied by. We then classify into sensor faults and unusual traffic conditions the STL anomalies of the "best" sensors with distance between them lower than 2000 m and whose produced data are correlated based on the correlation analysis of the data exploration step. Anomalies that are simultaneously detected from correlated traffic sensors are considered unusual traffic

conditions. This is based under the assumption that for proximal sensors, the unusual traffic condition will occur on the majority of sensors in the area. On the contrary, sensor faults are anomalies caused by sensor malfunctions, and hence, the adjacent traffic sensors will not capture them. Similar to the STL method, iForest is also applied to the "best" sensors' data to detect anomalies based on their counted vehicles. The fine-tuning of the created model includes tuning the contamination hyper-parameter that represents the proportion of outliers in data. The detected anomalies are classified to sensor faults and unusual traffic conditions similarly to the STL anomalies. We compare the classifications of the two methods and find the number of measurements that were detected as anomalies by both STL and iForest methods.

(5) Anomaly detection using multiple variables. This step further explores the quality of traffic data using six additional features to create the iForest model; the average speed of the vehicles and also five temporal features, namely, (i) the weekday, a binary variable showing whether the observation is recorded on a weekday (value 1) or weekend (value 0), (ii) daylight, a binary variable to show whether the measurement was made in day (value 1) or night (value 0), (iii) the day of the week of the measurement, a number in the range 0–6 (0 = Monday and 6 = Sunday), and (iv) the hour of the day of the measurement, a number in the range 0–23. It also employs the SHapley Additive exPlanation (SHAP) framework and, specifically, the TreeSHAP variant to explain the results of the created iForest model. Two types of visualizations are used: (i) SHAP summary plots that are beeswarm plots where each dot's position is determined by the feature on the y-axis and the Shapley value on the x-axis. The color represents the value of the feature from low to high, and (ii) SHAP dependence plots that show how a feature's value (x-axis) impacts the prediction (y-axis) of every sample (each dot) in a dataset and how the change in SHAP values across a feature's value range.

4. Case Vignettes

4.1. The Official Greek Open Government Data Portal

Data.gov.gr (accessed on 25 October 2022) is the official Greek data portal for Open Government Data (OGD). The latest version of the data portal was released in 2020 and provides access to 49 datasets published by the central government, local authorities, or other Greek public bodies classified in ten thematic areas, including environment, economy, and transportation.

The major update and innovation of the latest version of the Greek OGD portal was the introduction of an Application Programming Interface (API) that enables accessing and retrieving the data through either a graphical interface or code. The API is freely provided and can be employed to develop various products and services, including data intelligence applications. In order to use the API, users need to obtain a token by completing a registration process and providing personal information (i.e., name, email, and organization) as well as the reason for using the API.

The introduction of the API enables the timely provision of dynamic data that are frequently updated. The API can be used, for example, to retrieve datasets describing data related to a variety of transportation systems (e.g., road traffic for the Attica region, ticket validation of Attica's Urban Rail Transport, and route information and passenger counts of Greek shipping companies). The frequency of the data update varies.

4.2. Traffic Data in the Region of Attica

Traffic data for the Attica region in Greece are collected from traffic sensor nodes, which periodically transmit information regarding the number of vehicles on specific roads of Attica along with their speed. The data are aggregated hourly in order to avoid raising privacy issues. Data are updated hourly with only one hour delay.

We used the API provided by data.gov.gr (accessed on 25 October 2022) and collected 4,230,819 records for a 22-month period, i.e., from 5 November 2020 to 31 June 2022. Each record includes (a) the unique identifier of the sensor (e.g., MS834), (b) the road in which

the sensor is located along with (c) a detailed text description of its position, (d) the date and time of the measurement, (e) the absolute number of the vehicles detected by the sensor during the hour of measurement, and (f) their average speed in km per hour. The exact position of the sensor is a text description in the Greek language and usually provides details, including whether the sensor is located on a main or side road, or on an exit or entrance ramp, the direction of the road (e.g., direction to center), and the distance to main roads (e.g., "200 m from Kifisias avenue").

The collected traffic data have been produced by 425 sensors. We manually mapped the position of the sensors to latitude and longitude geographic coordinators in order to be able to present data in a map visualization. Specific position details are missing for one sensor (i.e., the sensor with identifier "MS339"), making it impossible to find its exact coordinates.

According to the data, the sensors did not start producing traffic data at the same time, and few of them stopped before the end of the period. Figure 1a presents the number of sensors that produced traffic data each month. Most sensors (370 or 87%) produce data from the first month of the data (November 2020), and then the number of sensors producing data gradually increases to reach the 420 sensors in June 2021. Specifically, during June 2021, 25 new sensors were introduced and, in the same month, two sensors stopped producing data. A new sensor also started producing data in July 2021, resulting again in 420 sensors. Finally, in December 2021, two new sensors started producing data, and two stopped keeping the number of sensors stable to 420 until the end of June 2022. We excluded from this work the data related to sensors that stopped producing data (namely sensors with IDs 'MS136', 'MS137', 'MS858', 'MS1000', and 'MS1001'). Finally, we gathered a total of 4,228,021 observations.



Figure 1. Number of sensors that produce traffic data per day and distribution of the IQR of the number of vehicles measured by each sensor. (a) Number of sensors that produce traffic data each month. (b) Distribution of the IQR of the vehicles measured by each sensor.

We then calculated the InterQuartile Range (IQR) of the counted vehicles measured by each sensor. Figure 1b shows the right-skewed distribution of IQR, meaning that few of the sensors probably counted large number of vehicles.

We also explored the correlation between all sensors by calculating the Pearson correlation coefficient for the number of vehicles counted each hour of the day by each sensor. In order for two sensors to be correlated, their distance should be lower than or equal to 2000 m. Based on this criterion, out of the 425 sensors, 417 have at least three correlated sensors (Table 1), while the average number of correlated sensors is 40.76. From these sensors, 404 have a Pearson correlation coefficient greater than 0.8, while 410 have a Pearson correlation coefficient greater than 0.7. Finally, there are six sensors that have one or two correlated sensors, while two sensors are isolated. **Table 1.** Correlation of sensors based on the counted number of vehicles counted each hour of the day. Correlated sensors have distance ≤ 2000 m and more than 3 correlations. Strongly correlated sensors have distance ≤ 2000 m and more than 3 correlations with Pearson coefficient ≥ 0.8 . Weakly correlated sensors have distance ≤ 2000 m and more than 3 correlations with Pearson coefficient ≥ 0.7 . Loosely correlated sensors have distance ≤ 2000 m and 1 or 2 correlations.

	Correlation
Total number of sensors	425
Correlated sensors	417
Strongly correlated sensors	404
Weakly correlated sensors	410
Loosely Correlated sensors	6
Isolated sensors	2

5. Missing Values

In this section, we search for observations that are missing from the traffic data based on two dimensions; (i) the time and (ii) the sensors. In the first case, we calculate the missing values per day, whilst in the second, the missing values per sensor.

Considering all the sensor measurements in the time frame between 5 November 2020 and 30 June 2022 and the time interval each of the 420 sensors was producing data, the number of the total potential observations would be 5,295,504. However, 1,067,483 observations (or 20.16%) are missing. Figure 2 presents the number of missing observations per day. The number of missing observations are increased until the end of May 2021. However, from June 2021, there is a significant decrease in the number of observations that are missing. Finally, the number of missing observations starts to eliminate after January 2022 and, specifically, after 2 January 2022. We select the time period from 2 January 2022 to 23 June 2022 as the "best" time period. This time period includes 75,425 missing observations.



Figure 2. Number of observations that are missing per day.

We also calculate the percent of missing observations for each sensor from 5 November 2020 to 30 June 2022. Figure 3 presents the street map with the 420 sensors and also the boxplot, which presents the distribution of the missing observations per sensor. The sensors are positioned on 93 main roads in the region of Attica. In Figure 3a, a mark is displayed over each sensor's longitude and latitude in the region of Attica. The colors of the marks indicate the percent of missing observations of the sensor; red color marks represent sensors with higher percentages of missing observations, green marks sensors with lower percent-

ages of missing observations, and yellow marks sensors with intermediate percentages of missing observations. In addition, the boxplot in Figure 3b presents the distribution of the number of missing observations per sensor. The median percent of missing observations is 33.1%, meaning that half of the sensors have less than or equal percentages of missing observations to the median, and half of the sensors have greater than or equal percentages of missing observations to it. In total, 50% of the sensors have a percentage of missing observations in the range of 15.5–33.43% (interquartile range box). In addition, according to the whiskers of the boxplot (bottom 25% and top 25% of the data values, excluding outliers), the percent of missing observations of each sensor may be as low as 10.3% and as high as 56.8%. In addition, based on our calculations, only eight sensors have less than 10% missing observations. Finally, only one sensor has a percentage of missing observations above 60%.



Figure 3. Map of sensors and boxplot for the missing values per sensor. (**a**) View of the 420 sensors on a map. Each point on the map represents a sensor. Red color marks represent sensors with higher percentages of missing values, green marks sensors with low percentages of missing values. (**b**) Percent of missing values per sensor in a boxplot showing the lower (Q1) and upper (Q3) quartile, the median and mean values. Data falling outside the lower (Q1)–upper (Q3) quartile range are plotted as outliers of the data.

6. Anomaly Detection

In this section, we detect the anomalies in the traffic data based on three anomaly detection methods, namely (i) flow-speed correlation, (ii) seasonal-trend decomposition with Loess, and (iii) isolation forest. The time window we select to perform the above analyses is the "best" time window starting from 2 January 2022 to 23 June 2022. In this time window, the dataset includes 1,679,898 records with measurements produced by 420 sensors.

6.1. Flow-Speed Correlation Analysis

We first calculate the number and percentages of anomalies per sensor based on the flow-speed correlation analysis method described in the research approach section (Section 3). In order to be able to calculate the number of vehicles that can pass in all lanes, we manually found the number of lanes that each sensor tracks and mapped them to the records. We discovered 1,230,928 observations that count more vehicles than the number calculated by the filter (59.4% of total potential observations). We also calculate

the number of anomalies per sensor. This number ranges from 0 to 3853 anomalies. In addition, the mean number of detected anomalies per sensor is 2937.8, while the median is 3264 anomalies per sensor. Table 2 presents the descriptive statistics for the anomalies detected per sensor.

The flow-speed correlation analysis showed that the vast majority of sensor recordings of the OGD dataset exceed the threshold value. For example, sensors MS792 and MS108 have 92.04% and 91.58% of anomalies, respectively. Only nine sensors have less than 10% anomalies based on the flow-speed correlation analysis (Table 3). We, henceforth, consider them as the "best" sensors for the selected time window. For these sensors, we calculate the number of anomalies based on the STL decomposition and iForest methods to further explore the quality of the traffic OGD. For both of these methods, identified missing values were imputed using the linear interpolation method.

Table 2. Descriptive statistics considering the number of anomalies per sensor using the flow-speed correlation analysis.

	Number of Anomalies	Mean Number of Anomalies
mean	2937.8	71.1
standard deviation	821.7	19.9
min	0	0
first quartile	2824	68.4
median	3264	79
third quartile	3416	82.75
max	3853	93.3

Table 3. Traffic sensors with less than 10% detected anomalies using the flow-speed correlation analysis.

Sensor ID	Hours of Anomaly	Percentage of Anomalies (%)
MS346	378	9.1
MS121	321	7.73
MS941	308	7.41
MS309	295	7.10
MS944	178	4.28
MS145	48	1.15
MS134	18	0.43
MS502	14	0.33
MS734	8	0.19

6.2. Anomaly Detection and Classification with STL Decomposition

The STL decomposition is applied to aggregated traffic data for the corresponding time window between 2 January 2022 and 23 June 2022 (4152 h). STL detects 4720 anomalies related to the nine "best" sensors (12.6% of 37,368 total records) of Table 3. Figure 4 shows an example of STL decomposition related to the observations of one sensor (sensor "MS734") for the selected time window. For anomaly detection, we examine the last curve of the STL method, the residual curve. Therefore, we apply the IQR method in the residual curve to specify the upper and lower fence for outlier detection. Observations above the upper limit and below the lower limit are considered anomalies. Finally, we set the scalar multiplied with IQR to three after a set of experiments. The upper and lower limits of the residual curve are defined using the following formula:

$$Upperlimit = Q_3 + 3IQR \tag{6}$$

$$Lowerlimit = Q_1 - 3IQR \tag{7}$$



Figure 4. STL decomposition of time series of sensor MS734.

The residual curve for sensor "MS734" depicted in Figure 5 shows that anomalies of this particular sensor are detected both in positive and negative peaks of the remainder component of STL.

Finally, we classify the STL traffic anomalies into sensor faults and unusual traffic conditions. We only classify anomalies detected by sensors with a distance between them lower than 2000 m and whose produced data are correlated based on the correlation analysis made in Section 4.2. Among the "best" sensors, only two pairs, i.e., "MS941" and "MS121", and "MS145" and "MS734", satisfy the previous criteria. Specifically, the distance between "MS941" and "MS121" is 1538.34 m, while the Pearson correlation between their produced data is 0.835128. In addition, the distance between "MS145" and "MS734" is 1333.49 m, while the Pearson correlation between their produced data is 0.783.

Table 4 shows the total number of anomalies detected by the STL decomposition method for the nine reliable sensors and the number of STL anomalies classified as unusual traffic conditions and sensor faults. The classification shows that unusual traffic conditions and sensor faults are almost equally distributed within the total number of anomalies. We further explored the classified anomalies. In total, 97% of unusual traffic conditions were detected during the day (between 07:00 and 20:00) and 56% were classified on weekdays. In total, 90% of sensor faults were detected in daylight, while 62% were detected on weekdays.

Sensor ID	STL Anomalies	Unusual Traffic Conditions	Sensor Faults
MS121	359	179	180
MS134	502	n/a	n/a
MS145	464	256	208
MS309	679	n/a	n/a
MS346	489	n/a	n/a
MS502	384	n/a	n/a
MS734	615	256	359
MS941	454	179	275
MS944	774	n/a	n/a

Table 4. Anomalies detected by the STL decomposition.

Figure 6 depicts the anomalies detected in the data produced by sensor "MS734" on the residual curve upon the original time series. Among these anomalies, unusual traffic conditions are depicted as green dots. These anomalies are also detected by STL on the correlated sensor "MS145". On the contrary, anomalies that are not recorded by proximal sensors on the same timestamp are considered sensor faults and depicted with red dots.



Figure 5. Residual curve of sensor "MS734" with the upper and lower fences of the IQR method. The green dots represent unusual traffic conditions. Red dots represent sensor faults.



Figure 6. Time series of sensor "MS734" with anomalies detected by STL decomposition. Red dots are anomalies classified as "sensor faults", and green dots are anomalies classified as "unusual traffic conditions".

6.3. Anomaly Detection and Classification with Isolation Forest

Next, we apply the isolation Forest (iForest) algorithm to the OGD traffic dataset. Similar to the STL method, iForest is implemented on the same time window between 2 January 2022 and 23 June 2022. The algorithm is deployed using python's Scikit-learn machine learning library (https://scikit-learn.org/stable/modules/generated/sklearn. ensemble.IsolationForest.html/ accessed on 25 October 2022). Furthermore, the only hyper-parameter of the model needed to be tuned is the contamination parameter, which represents the proportion of outliers in the dataset. Similar to other ensemble learning techniques, iForest is made up of a large number of decision trees. The number of isolation trees is represented by the parameter $m_{estimators}$, which is set to 100 for all the time series in this experiment. Finally, the parameter $max_{samples}$, which is the number of random samples from the original dataset that will be created for the isolation trees, is set to 256. The iForest algorithm detects 4353 anomalies out of 37,368 records (11.6%) of the data from the nine sensors.

Similar to the STL method, we apply the classification method to the data of the correlated sensors "MS121", "MS145", and "MS734" and "MS941". Table 5 shows the total number of anomalies (out of 4152 records per sensor) that were identified by iForest

and the classified anomalies among the correlated sensors. The majority of anomalies are classified as sensor faults. Especially for sensors "MS121" and "MS941", 78.7% and 89.7% of the detected anomalies, respectively, are classified as sensor faults. We further explored the classified anomalies. A vast majority of classified unusual traffic conditions, almost 98%, were detected during daylight (between 07:00 and 20:00) and 55% of them during weekdays. Finally, 81% of the detected sensor faults were recorded both in daylight hours and on weekdays.

Sensor ID	iForest Anomalies	Unusual Traffic Conditions	Sensor Faults
MS121	287	61	226
MS134	441	n/a	n/a
MS145	494	214	280
MS309	619	n/a	n/a
MS346	535	n/a	n/a
MS502	316	n/a	n/a
MS734	575	214	361
MS941	385	61	324
MS944	701	n/a	n/a

Table 5. Anomalies detected by the iForest algorithm.

Figure 7 displays the anomalies detected by iForest classified as unusual traffic conditions (green dots) and as sensor faults (red dots) for sensor "MS734".



Figure 7. Time series of sensor "MS734" with anomalies detected by the iForest algorithm. Red dots are anomalies classified as "sensor faults" and green dots are anomalies classified as "unusual traffic conditions".

In Table 6, the anomalies detected by STL and iForest are compared. For the nine "best" sensors, each column points out the (i) total overlaps, (ii) overlaps of unusual traffic conditions, and (iii) overlaps in sensor faults of the anomalies detected by the two methods. We observe that only sensor "MS944" has a high value of overlap, with 420 anomalies detected by both anomaly detection algorithms. On the contrary, our methods detect only 65 and 88 common anomalies for sensors "MS121" and "MS941", respectively. According to this table, the overlaps regarding the classification of anomalies are very low between the two methods. Only sensor MS734 has 111 overlaps on sensor faults. The major differences in the two proposed methods are also depicted in Figures 6 and 7. The iForest technique detects anomalies mainly on the low and high peaks of the time series since the algorithm manages to classify anomalies due to their significant deviation from the normal pattern. On the contrary, as Figure 6 illustrates, STL applies anomaly detection on the residual curve with the IQR method thinking of the time series as a combination of trend, seasonality and remainder. Thus, it highlights anomalies taking into consideration trend and seasonality

effects of the time series. These might be significant reasons for the small number of overlaps between the two methods, as Table 6 suggests. However, our experiments are conducted only on 9 out of 420 sensors of the traffic network due to the flow-speed correlation analysis results. Thus, we need a more reliable dataset with a greater number of sensors to correctly compare the two methods.

Sensor ID	Total Overlap	Unusual Traffic Conditions	Sensor Faults
MS121	88	15	52
MS134	222	n/a	n/a
MS145	115	34	45
MS309	180	n/a	n/a
MS346	177	n/a	n/a
MS502	176	n/a	n/a
MS734	211	34	111
MS941	65	15	29
MS944	420	n/a	n/a

Table 6. Comparison of STL decomposition and isolation forest anomaly detection algorithms.

7. Anomaly Detection Using Multiple Variables

In order to further explore the quality of the OGD traffic dataset, we train the isolation Forest (iForest) algorithm on additional features and make explanations using the SHapley Additive exPlanation (SHAP) framework to interpret the classification of anomalies. Our goal is to explore specific patterns among the traffic dataset that may lead to anomalies and, hence, deteriorate its quality. In this direction, the following explanations do not interpret the exact reasons in anomalies but rather give an estimation of what the indicators are that might contribute to a traffic anomaly. The iForest algorithm is suitable for highdimensional anomaly detection involving multiple features during the partitioning process. The main difference from the one-feature implementation is that the algorithm randomly selects a feature every time it partitions a data point. Then the model selects a split value between the minimum and maximum values of the selected feature. For that reason, we calculated four temporal features from our time series: (i) the weekday, a binary variable showing whether the observation is recorded on a weekday (value 1) or weekend (value 0), (ii) daylight, a binary variable to show whether the measurement was made in the day (value 1) or night (value 0), (iii) the day of the week of the measurement, a number in the range 0-6 (0 = Monday and 6 = Sunday), and (iv) the hour of the day of the measurement, a number in the range 0–23. Finally, we also add the average speed of vehicles measured by the traffic sensors for one hour.

The iForest algorithm is explained using the SHAP values of every data instance obtained by the nine "best" sensors. For the calculation of SHAP values, the TreeSHAP model [55] was used to provide an anomaly explanation.

In order to understand how several variables affect the detection of anomalies, we created a SHAP summary plot. Figures 8 and 9 show the global interpretation of Shapley values for sensors MS734 and MS941, respectively. The plots present the features with the highest impact on the output of the model in descending global importance. The X-axis shows the Shapley values of each measurement for every feature; Shapley values below zero mean that the model classifies a data instance as an anomaly, while Shapley values greater than zero mean that iForest classifies a data instance (every observation). Each dot is also colored by the value of that feature from high to low, with red dots representing high values and blue dots as low values of a feature. According to the summary plots in Figures 8 and 9, "WeekDay" and "Average Speed" are the two features with the highest impact for anomaly detection. "WeekDay" is a feature that indicates whether the measurement took place on a workday "WeekDay" = 1) or during the weekend ("WeekDay" = 0). Therefore, high values of this feature indicate that the recording happened on a weekday. Data points

that are recorded on weekends are represented in the summary plots with blue color and have low (negative) Shapley values, meaning that anomalies are more likely detected on weekends. On the contrary, data instances that are recorded in a weekday are presented in the summary plots with red color and have high (positive) Shapley values, indicating that normal points are more likely detected on a weekday. We are also interested in long left tails that form the beeswarm plots because they correspond to extreme anomalies (extremely negative Shapley values). In the case of sensor MS734 (Figure 8), extremely low values of "Average Speed" correspond to very low Shapley values meaning that low average speeds are indicators of extreme anomalies. Moreover, in the same plot, high values of "Counted vehicles", a feature with low global importance, imply the existence of very high anomalies for both sensors. In the same way, both summary plots illustrate that blue dots of variable "daylight" are mostly gathered on the negative part of X-axis, implying that the algorithm detects anomalies during night hours. Finally, "Day of the Week" does not display a clear pattern regarding its influence on the model's outcome.



Figure 8. Distribution of SHAP values for each feature, sensor "MS734".



Figure 9. Distribution of SHAP values for each feature, sensor "MS941".

In order to fully understand the relationship between a feature's values and the model's outcomes and estimate the influence of variables on anomaly detection, we also create SHAP dependence plots. A dependence plot depicts every data instance (i.e., every row) as a blue dot. These dots form a scatter plot of the feature's raw values versus the corresponding SHAP values. Figure 10 illustrates dependence plots for the two measured variables of the traffic sensors: average speed and counted vehicles for sensor MS734. SHAP values above the y = 0 line lead to the detection of normal points, whereas those below it are considered anomalies. For instance, the pattern of the Figure 10a plot indicates that sensor measurements with zero average speed are considered anomalies. Furthermore, there is a range of average speed between 33 and 42 km/h, where the corresponding SHAP values are positive, indicating that sensor measurements between this range are more likely

considered normal. Figure 10b shows the number of counted vehicles measured by sensor MS734 and their corresponding SHAP values. For this sensor, measurements with more than 3800 vehicles per hour have a significant decrease in SHAP values, clearly indicating anomalous traffic behavior. Moreover, recordings with counted vehicles between zero and 3800 vehicles per hour can be both normal and anomalous.



Figure 10. SHAP dependence plots of sensor "MS734". (a) Average speed vs. SHAP values. (b) Traffic flow vs. SHAP values.

8. Discussion

During the last two decades, the Open Government Data (OGD) phenomenon has rapidly evolved. In these years, numerous OGD portals have already opened up their data for free reuse, and the economic and social potential of OGD has significantly increased. In order to facilitate the involvement of OGD in the creation of data intelligence applications, various technologies have been used. Linked data is a technological paradigm that was proposed early on [1] and has already been adopted by many OGD portals developed by, e.g., the Scottish Government (https://statistics.gov.scot/ accessed on 25 October 2022) and the UK Department for Communities and Local Government (DCLG) (http:// opendatacommunities.org/ accessed on 25 October 2022). Linked data not only facilitated the integration of data both inside and across data portals but also ensured the provision of high-quality data. This is of vital importance in specific types of OGD, such as statistical data, where data are described in different granularity levels [59]. The provision of highquality statistical data that can be easily integrated enables the emergence of added-value scenarios that can be easily implemented. Indeed, in our previous work, we employed linked statistical data from the Scottish OGD portal to facilitate policy-making related to house prices in Scotland using machine learning and explainable artificial intelligence technologies [20].

Today, however, the amount of Open Government Data (OGD) that are being produced and disseminated is exponentially growing while new types of data are also being generated, such as dynamic OGD (e.g., traffic data generated by sensors), which were recognized as an important part of OGD. At the same time, the recent emergence of innovative data analysis and exploitation methods (e.g., artificial intelligence including machine learning) opens up new opportunities for exploiting OGD. It is, hence, important to define and address the new challenges that are related to the exploitation of these new types of data using innovative methods in order to extract the potential that previous types of data and methods cannot.

In this context, this paper focused on traffic data, which is a prominent example of high-value data, and explored the quality of Attica traffic data. To the authors' knowledge, this is the first time a study has explored the quality of OGD by studying their data and not the metadata provided by OGD portals. For example, previous works are based on assessing metadata, such as the data formats available (including whether they are

provided in a machine readable format) [4,6], the accuracy of the metadata (i.e., whether the provided metadata are correct) [4,60], the timeliness of the metadata (i.e., whether metadata are up-to-date) [61], and the discoverability, reusability and accessibility of the datasets [4,62].

The data used in this work are provided by data.gov.gr (accessed on 25 October 2022)), the official Greek OGD portal, through an API, ensuring their immediate availability and are updated hourly with only a one hour delay. The research approach includes steps that can applied to assess the quality of more traffic datasets provided by other OGD portals. We found that considering the time frame between 5 November 2020 and 30 June 2022, and the number of days each one of the 420 sensors was producing data, 20.16% of the observations are missing. In addition, 50% of the sensors have a percentage of missing values in the range 15.5–33.43% (interquartile range box). We also used flow-speed correlation analysis to detect anomalous flow-speed correlations and two methods, namely Seasonal-Trend decomposition using Loess (STL) and unsupervised learning isolation Forest (iForest), to detect deviations from the normal traffic pattern. The flow-speed correlation analysis found that the mean percent of anomalies per sensor is 71.1%. Only nine sensors have less than 10% anomalies based on the flow-speed correlation analysis. For these nine sensors, STL decomposition detected 4720 anomalies (12.6%) and isolation Forest, 4353 anomalies (11.6%). However, the comparison of the two methods showed that there a few overlaps between the proposed methods. STL manages to detect anomalies by taking into consideration seasonality and trend effects, while iForest mainly detects the high and low peaks of the corresponding time series. As a result, using both methods to detect anomalies appears to be a good strategy. Furthermore, our proposed classification method showed that both unusual traffic conditions and sensor faults are detected mainly on weekdays and during daylight hours, while the iForest algorithm manages to detect more sensor faults than the STL method. Finally, in order to have an estimation of the factors that influence traffic anomalies, we implemented the SHAP framework upon the iForest model. The results showed that average speed and the day during the recording (weekday or weekend) are the most significant factors. Moreover, we found that very low values of average speed have a high contribution to the identification of anomalies, while sensor measurements with counted vehicles per hour greater than 3800 lead to the classification of extreme anomalies.

9. Conclusions

The findings of this study demonstrate that, while traffic data from the Greek OGD portal may be retrieved promptly via the API and are constantly updated, they confront major quality difficulties. However, this may not be the case for all OGD portals' dynamic data. We believe that further research in the dynamic data of other OGD sites will disclose datasets of higher quality that might be potentially used in added-value scenarios. In any case, the exploration of big and real-time data provided by OGD portals will enable identifying and addressing organizational and technical challenges that hamper the effective dissemination of high-value government data.

Author Contributions: Conceptualization, E.K. and A.K.; methodology, E.K.; software, A.K. and P.B.; data curation, A.K. and P.B.; writing—original draft, A.K. and P.B.; writing—review and editing, A.K. and E.K.; supervision, E.K. and K.T.; Project administration, E.K. and K.T. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Publicly available datasets were analyzed in this study. These data can be found here: (https://data.gov.gr/datasets/road_traffic_attica/) (accessed on 25 October 2022).

Conflicts of Interest: The authors declare no conflict of interest.

Abbreviations

The following abbreviations are used in this manuscript:

- API Application Programming Interface
- HVD High-Value Datasets
- OGD Open Government Data
- STL Seasonal-Trend decomposition using Loess

References

- 1. Kalampokis, E.; Tambouris, E.; Tarabanis, K. A classification scheme for open government data: Towards linking decentralised data. *Int. J. Web Eng. Technol.* **2011**, *6*, 266–285. [CrossRef]
- 2. Cox, P.; Alemanno, G. Directive 2003/98/EC of the European Parliament and of the council of 17 November 2003 on the re-use of public sector information. *Off. J. Eur. Union* **2003**, *46*, 1–156.
- 3. Peled, A. When transparency and collaboration collide: The USA Open Data program. *J. Am. Soc. Inf. Sci. Technol.* 2011, 62, 2085–2094. [CrossRef]
- Attard, J.; Orlandi, F.; Scerri, S.; Auer, S. A systematic review of open government data initiatives. *Gov. Inf. Q.* 2015, 32, 399–418. [CrossRef]
- Shadbolt, N.; O'Hara, K.; Berners-Lee, T.; Gibbins, N.; Glaser, H.; Hall, W.; schraefel, M. Linked Open Government Data: Lessons from Data.gov.uk. *IEEE Intell. Syst.* 2012, 27, 16–24. [CrossRef]
- 6. Nikiforova, A.; McBride, K. Open government data portal usability: A user-centred usability analysis of 41 open government data portals. *Telemat. Inform.* 2021, *58*, 101539. [CrossRef]
- Lourenço, R.P. An analysis of open government portals: A perspective of transparency for accountability. *Gov. Inf. Q.* 2015, 32, 323–332. [CrossRef]
- 8. Ruijer, E.; Grimmelikhuijsen, S.; Meijer, A. Open data for democracy: Developing a theoretical framework for open data use. *Gov. Inf. Q.* **2017**, *34*, 45–52. [CrossRef]
- Yang, Z.; Ha, S.; Kankanhalli, A.; Um, S. Understanding the determinants of the intention to innovate with open government data among potential commercial innovators: A risk perspective. *Internet Res.* 2022, *ahead-of-print*. [CrossRef]
- 10. Leviäkangas, P.; Molarius, R. Open government data policy and value added—Evidence on transport safety agency case. *Technol. Soc.* **2020**, *63*, 101389. [CrossRef]
- 11. Janssen, M.; Charalabidis, Y.; Zuiderwijk, A. Benefits, adoption barriers and myths of open data and open government. *Inf. Syst. Manag.* **2012**, *29*, 258–268. [CrossRef]
- 12. Zhang, J.; Puron-Cid, G.; Gil-Garcia, J.R. Creating public value through Open Government: Perspectives, experiences and applications. *Inf. Polity* 2015, 20, 97–101. [CrossRef]
- Tambouris, E.; Tarabanis, K. Towards Inclusive Integrated Public Service (IPS) Co-Creation and Provision. In Proceedings of the DG.O2021: The 22nd Annual International Conference on Digital Government Research, Omaha, NE, USA, 9–11 June 2021; Association for Computing Machinery: New York, NY, USA, 2021; pp. 458–462. [CrossRef]
- 14. Wang, H.J.; Lo, J. Adoption of open government data among government agencies. Gov. Inf. Q. 2016, 33, 80–88. [CrossRef]
- 15. Parliament, E.; Council, E. Directive (EU) 2019/1024 of the European Parliament and of the Council of 20 June 2019 on open data and the re-use of public sector information (recast). *Off. J. Eur. Union* 2019, 172, 56–83.
- 16. Teh, H.Y.; Kempa-Liehr, A.W.; Wang, K.I.K. Sensor data quality: A systematic review. J. Big Data 2020, 7, 11. [CrossRef]
- 17. Nikiforova, A.; Lnenicka, M. A multi-perspective knowledge-driven approach for analysis of the demand side of the Open Government Data portal. *Gov. Inf. Q.* 2021, *38*, 101622. [CrossRef]
- 18. Lnenicka, M.; Nikiforova, A. Transparency-by-design: What is the role of open data portals? *Telemat. Inform.* **2021**, *61*, 101605. [CrossRef]
- Kalampokis, E.; Karacapilidis, N.; Tsakalidis, D.; Tarabanis, K. Artificial Intelligence and Blockchain Technologies in the Public Sector: A Research Projects Perspective. In *Electronic Government*; Janssen, M., Csáki, C., Lindgren, I., Loukis, E., Melin, U., Viale Pereira, G., Rodríguez Bolívar, M.P., Tambouris, E., Eds.; Springer International Publishing: Cham, Swizerland, 2022; pp. 323–335.
- Karamanou, A.; Kalampokis, E.; Tarabanis, K. Linked Open Government Data to Predict and Explain House Prices: The Case of Scottish Statistics Portal. *Big Data Res.* 2022, *30*, 100355. [CrossRef]
- 21. Jiang, W.; Luo, J. Graph neural network for traffic forecasting: A survey. Expert Syst. Appl. 2022, 207, 117921. [CrossRef]
- 22. Kalampokis, E.; Tambouris, E.; Tarabanis, K. Open Government Data: A Stage Model. In *Electronic Government*; Janssen, M., Scholl, H.J., Wimmer, M.A., Tan, Y.H., Eds.; Springer: Berlin/Heidelberg, Germany, 2011; pp. 235–246.
- 23. Ubaldi, B. Open Government Data; OECD Publishing: Paris, France, 2013. [CrossRef]
- 24. Sołtysik-Piorunkiewicz, A.; Zdonek, I. How Society 5.0 and Industry 4.0 Ideas Shape the Open Data Performance Expectancy. *Sustainability* 2021, 13, 917. [CrossRef]
- Pereira, G.V.; Macadar, M.A.; Luciano, E.M.; Testa, M.G. Delivering public value through open government data initiatives in a Smart City context. *Inf. Syst. Front.* 2017, 19, 213–229. [CrossRef]

- 26. Nikiforova, A. Smarter Open Government Data for Society 5.0: Are your open data smart enough? *Sensors* 2021, 21, 5204. [CrossRef] [PubMed]
- 27. Aggarwal, C.C. An Introduction to Outlier Analysis. In Outlier Analysis; Springer: New York, NY, USA, 2013; pp. 1–40. [CrossRef]
- 28. Chandola, V.; Banerjee, A.; Kumar, V. Anomaly Detection: A Survey. ACM Comput. Surv. 2009, 41, 15. [CrossRef]
- 29. Braei, M.; Wagner, S. Anomaly Detection in Univariate Time-series: A Survey on the State-of-the-Art. arXiv 2020. [CrossRef]
- Muruti, G.; Rahim, F.A.; bin Ibrahim, Z.A. A Survey on Anomalies Detection Techniques and Measurement Methods. In Proceedings of the 2018 IEEE Conference on Application, Information and Network Security (AINS), Langkawi, Malaysia, 21–22 November 2018; pp. 81–86. [CrossRef]
- 31. Kumar, V.; Banerjee, A.; Chandola, V. *Anomaly Detection for Symbolic Sequences and Time Series Data*; University of Minnesota: Minneapolis, MN, USA, 2009.
- Chauhan, S.; Vig, L. Anomaly detection in ECG time signals via deep long short-term memory networks. In Proceedings of the 2015 IEEE International Conference on Data Science and Advanced Analytics (DSAA), Paris, France, 19–21 October 2015; pp. 1–7. [CrossRef]
- 33. Liu, C.; Li, Y.; Zhou, G.; Shen, W. A sensor fusion and support vector machine based approach for recognition of complex machining conditions. *J. Intell. Manuf.* 2018, 29, 1739–1752. [CrossRef]
- Zhang, C.; Song, D.; Chen, Y.; Feng, X.; Lumezanu, C.; Cheng, W.; Ni, J.; Zong, B.; Chen, H.; Chawla, N.V. A Deep Neural Network for Unsupervised Anomaly Detection and Diagnosis in Multivariate Time Series Data. *Proc. AAAI Conf. Artif. Intell.* 2019, 33, 1409–1416. [CrossRef]
- 35. Bachechi, C.; Rollo, F.; Po, L. Detection and classification of sensor anomalies for simulating urban traffic scenarios. *Clust. Comput.* **2022**, 25, 2793–2817. [CrossRef]
- 36. Cleveland, R.B.; Cleveland, W.S.; McRae, J.E.; Terpenning, I. STL: A seasonal-trend decomposition. J. Off. Stat. 1990, 6, 3–73.
- Yang, L.; Wen, Q.; Yang, B.; Sun, L. A Robust and Efficient Multi-Scale Seasonal-Trend Decomposition. In Proceedings of the ICASSP 2021—2021 IEEE International Conference on Acoustics, Speech and Signal Processing (ICASSP), Toronto, ON, Canada, 6–11 June 2021; pp. 5085–5089. [CrossRef]
- Wen, Q.; Gao, J.; Song, X.; Sun, L.; Xu, H.; Zhu, S. RobustSTL: A Robust Seasonal-Trend Decomposition Algorithm for Long Time Series. Proc. AAAI Conf. Artif. Intell. 2019, 33, 5409–5416. [CrossRef]
- Wen, Q.; Zhang, Z.; Li, Y.; Sun, L. Fast RobustSTL: Efficient and Robust Seasonal-Trend Decomposition for Time Series with Complex Patterns. In Proceedings of the 26th ACM SIGKDD International Conference on Knowledge Discovery & Data Mining, Virtual Event, 23–27 August 2020.
- Bachechi, C.; Rollo, F.; Po, L. Real-Time Data Cleaning in Traffic Sensor Networks. In Proceedings of the 2020 IEEE/ACS 17th International Conference on Computer Systems and Applications (AICCSA), Antalya, Turkey, 2–5 November 2020; pp. 1–8. [CrossRef]
- 41. Hu, J.; Zhu, E.; Wang, S.; Liu, X.; Guo, X.; Yin, J. An Efficient and Robust Unsupervised Anomaly Detection Method Using Ensemble Random Projection in Surveillance Videos. *Sensors* **2019**, *19*, 4145. [CrossRef]
- 42. Ma, N.; Peng, Y.; Wang, S.; Leong, P.H.W. An Unsupervised Deep Hyperspectral Anomaly Detector. *Sensors* 2018, 18, 693. [CrossRef]
- Gouda, W.; Tahir, S.; Alanazi, S.; Almufareh, M.; Alwakid, G. Unsupervised Outlier Detection in IOT Using Deep VAE. Sensors 2022, 22, 6617. [CrossRef] [PubMed]
- Almudévar, A.; Sevillano, P.; Vicente, L.; Preciado-Garbayo, J.; Ortega, A. Unsupervised Anomaly Detection Applied to Φ-OTDR. Sensors 2022, 22, 6515. [CrossRef] [PubMed]
- 45. Liu, F.T.; Ting, K.M.; Zhou, Z.H. Isolation Forest. In Proceedings of the 2008 Eighth IEEE International Conference on Data Mining, Pisa, Italy, 15–19 December 2008; pp. 413–422. [CrossRef]
- Lesouple, J.; Baudoin, C.; Spigai, M.; Tourneret, J.Y. Generalized isolation forest for anomaly detection. *Pattern Recognit. Lett.* 2021, 149, 109–119. [CrossRef]
- 47. Ding, Z.; Fei, M. An Anomaly Detection Approach Based on Isolation Forest Algorithm for Streaming Data using Sliding Window. *IFAC Proc. Vol.* **2013**, *46*, 12–17. [CrossRef]
- van de Wiel, L.; van Es, D.M.; Feelders, A.J. Real-Time Outlier Detection in Time Series Data of Water Sensors. In Advanced Analytics and Learning on Temporal Data: 5th ECML PKDD Workshop, AALTD 2020, Ghent, Belgium, September 18, 2020, Revised Selected Papers; Springer: Berlin/Heidelberg, Germany, 2020; pp. 155–170. [CrossRef]
- 49. Rudin, C. Stop explaining black box machine learning models for high stakes decisions and use interpretable models instead. *Nat. Mach. Intell.* **2019**, *1*, 206–215. [CrossRef]
- Laios, A.; Kalampokis, E.; Johnson, R.; Thangavelu, A.; Tarabanis, C.; Nugent, D.; De Jong, D. Explainable Artificial Intelligence for Prediction of Complete Surgical Cytoreduction in Advanced-Stage Epithelial Ovarian Cancer. *J. Pers. Med.* 2022, *12*, 607. [CrossRef]
- Petsis, S.; Karamanou, A.; Kalampokis, E.; Tarabanis, K. Forecasting and explaining emergency department visits in a public hospital. *J. Intell. Inf. Syst.* 2022, 59, 479–500. [CrossRef]
- 52. Parsa, A.B.; Movahedi, A.; Taghipour, H.; Derrible, S.; Mohammadian, A.K. Toward safer highways, application of XGBoost and SHAP for real-time accident detection and feature analysis. *Accid. Anal. Prev.* **2020**, *136*, 105405. [CrossRef]

- 53. Sejr, J.H.; Schneider-Kamp, A. Explainable outlier detection: What, for Whom and Why? *Mach. Learn. Appl.* **2021**, *6*, 100172. [CrossRef]
- Ribeiro, M.T.; Singh, S.; Guestrin, C. "Why Should I Trust You?": Explaining the Predictions of Any Classifier. In Proceedings of the 22nd ACM SIGKDD International Conference on Knowledge Discovery and Data Mining, San Francisco, CA, USA, 13–17 August 2016; Association for Computing Machinery: New York, NY, USA, 2016; pp. 1135–1144. [CrossRef]
- 55. Lundberg, S.; Lee, S.I. A Unified Approach to Interpreting Model Predictions. arXiv 2017. [CrossRef]
- Choi, H.; Kim, D.; Kim, J.; Kim, J.; Kang, P. Explainable anomaly detection framework for predictive maintenance in manufacturing systems. *Appl. Soft Comput.* 2022, 125, 109147. [CrossRef]
- 57. Antwarg, L.; Miller, R.M.; Shapira, B.; Rokach, L. Explaining anomalies detected by autoencoders using Shapley Additive Explanations. *Expert Syst. Appl.* **2021**, *186*, 115736. [CrossRef]
- Kim, D.; Antariksa, G.; Handayani, M.P.; Lee, S.; Lee, J. Explainable Anomaly Detection Framework for Maritime Main Engine Sensor Data. Sensors 2021, 21, 5200. [CrossRef] [PubMed]
- Karamanou, A.; Kalampokis, E.; Tarabanis, K. Integrated Statistical Indicators from Scottish Linked Open Government Data. Data Brief 2022, 46, 108779. [CrossRef] [PubMed]
- Reiche, K.J.; Höfig, E. Implementation of metadata quality metrics and application on public government data. In Proceedings of the 2013 IEEE 37th Annual Computer Software and Applications Conference Workshops, Kyoto, Japan, 22–26 July 2013; pp. 236–241.
- Kučera, J.; Chlapek, D.; Nečaský, M. Open Government Data Catalogs: Current Approaches and Quality Perspective. In Technology-Enabled Innovation for Democracy, Government and Governance; Kő, A., Leitner, C., Leitold, H., Prosser, A., Eds.; Springer: Berlin/Heidelberg, Germany, 2013; pp. 152–166.
- 62. Máchová, R.; Hub, M.; Lnenicka, M. Usability evaluation of open data portals. Aslib J. Inf. Manag. 2018, 70, 252–268. [CrossRef]